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Correlations for the interfacial terms in the fluid dissipation rate equation and Reynolds
stress equations are established for particle-laden flows, based on data from the
interfaced-resolved direct numerical simulations of particle sedimentation in a periodic
domain at a density ratio ranging from 0.01 to 1000, a particle concentration ranging
from 2.3 % to 30.2 % and a particle Reynolds number below 250. The correlations for
the mean drag and the pseudo-turbulent kinetic energy are also reported, which are
used for the modelling of the interfacial term in the fluid dissipation rate equation. The
interfacial term correlations obtained are then incorporated in the Reynolds stress model
(RSM) (i.e. second-moment closure) for the simulation of vertical turbulent channel flows
laden with the finite-size particles at relatively low particle volume fractions. The results
show that the RSM with new interfacial term correlations can quantitatively predict
particle-induced turbulence enhancement or suppression in vertical channel flows.
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1. Introduction

Turbulent dispersed multiphase flows are encountered in many processes in nature and
industry, such as fluidized beds, drug delivery, drilling process and sand storms. The
two-fluid model (i.e. the Euler—Euler method) (Anderson & Jackson 1967; Ding &
Gidaspow 1990; Ishii & Hibiki 2010) and the discrete particle (or element) model
(i.e. the Euler—Lagrange method) (Tsuji, Kawaguchi & Tanaka 1993; Deen et al. 2007,
Zhu et al. 2007) are two models widely used for particle-laden flows in engineering
applications because of much lower computational resource requirements compared with
direct numerical simulations (DNS). For the two-fluid model, modelling of the interphase
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Politano et al. Troshko & Ma et al. Yao & Rzehak &
(2003) Hassan (2001) (2017) Morel (2004) Krepper (2013)
Length scale B2/ dp dy A~d, K2/ or d,
Velocity scale Vk Uy Uy (e)'/3 uy or vk
Time scale t k/e dy/u, dy/uy d§/3/81/3 K372/ (guy) or dp/\/E
Ce3 1.92 1.35Cp 0.3Cp 1.0 1.0

Table 1. Summary of characteristic scales for the modelling of the dissipation interfacial term based on
dimensional analysis.

momentum and energy transfers is crucial for accurate simulations of industrial-scale
multiphase flows (Crowe, Troutt & Chung 1996; Crowe et al. 2011; Joshi & Nandakumar
2015).

The turbulence models for particle-laden flows have been established based on the
modification of single-phase turbulence models. In the past two decades numerous
studies have been devoted to developing the two-equation multiphase turbulence models
(including k—¢ and k—w) with transport equations for the turbulent kinetic energy (TKE)
and dissipation rate (Sato & Hishida 1996; Troshko & Hassan 2001; Politano, Carrica &
Converti 2003; Rzehak & Krepper 2013; Ma et al. 2017). Sato, Sadatomi & Sekoguchi
(1981) proposed a particle-induced turbulence viscosity. The effects of particles on the
turbulence can be captured through the inclusion of the interfacial (source) terms created
by particles in the Reynolds stress and dissipation rate equations (Troshko & Hassan 2001;
Politano et al. 2003; Rzehak & Krepper 2013; Fox 2014; Ma et al. 2017, 2020b). For
the particle-induced turbulence, the interfacial term in the TKE equation, I, is generally
modelled as the product of the interphase mean drag force per unit volume and the
interphase mean slip velocity (Troshko & Hassan 2001; Colombo & Fairweather 2015;
Mehrabadi ef al. 2015; Ma et al. 2017).

The model of the interfacial term in the dissipation rate equation, /., is much more
complex. Based on dimensional analysis, a general expression of I, reads

I, = Ce3—, (1.1)
T

where t represents a characteristic time scale of the flow and C,3 is an empirical
coefficient. The determination of the characteristic scales for particle-laden flows is
still a subject of debate in the literature because of the intrinsic multiscale property of
particle-laden flows. Some typical choices of the characteristic scales and the coefficient
C,3 are summarized in table 1. For shear-dominated turbulence, the large-scale eddy time
T = k/e is normally chosen as the characteristic time scale (Politano et al. 2003). For
purely particle-induced pseudo-turbulence (i.e. the turbulence generated by the particles),
the characteristic length scale is related to the wake length L,, or particle diameter d,,.
Therefore, the time scale can be selected as T = d,, /u, (Troshko & Hassan 2001; Ma et al.
2017), where u, is the particle slip velocity. Risso ez al. (2008) suggested that L,, < d,,/Cp,
while Du Cluzeau, Bois & Toutant (2019) argued that the wake length is in the range of
2d,, ~ 4d,,. For the situation where both shear-induced turbulence and particle-induced
turbulence are important, several hybrid scales have been proposed in the literature (Yao
& Morel 2004; Rzehak & Krepper 2013).

The model of the dissipation rate interfacial term can also be derived from
fundamentals. Schwarzkopf, Crowe & Dutta (2009) derived a dissipation rate transport
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equation for particle-laden flows, and proposed a model of the dissipation rate interfacial
term based on the point-particle approximation as

=Cayy ka u” ’Ik (12)

where f} is the drag factor, u; and uy,; are the local fluid and particle velocities, respectively,
and C.; is an empirical coefficient. Experimental data on particle sedimentation in a
quiescent fluid were used to obtain the coefficient C. for solid volume fractions between
10~ and 1072 and particle Reynolds numbers between 30 and 1000. The primary purpose
of tlzle present work is to provide an extending model for the solid volume fraction above
107~

Although the two-equation models have been widely used to simulate multiphase flows,
they fail to account for the strong anisotropy of particle-induced velocity fluctuations
(Mehrabadi et al. 2015; Ma et al. 2020b). This limitation can be overcome by employing
the Reynolds stress model (RSM) (Cokljat et al. 2006; Masood, Rauh & Delgado 2014;
Capecelatro, Desjardins & Fox 2016a,b; Riella, Kahraman & Tabor 2019; Baker et al.
2020; Ma, Lucas & Bragg 2020a; Ma et al. 2020b; Shi & Rzehak 2020). For weakly
inertial particle-laden flows, Capecelatro et al. (20160) developed a multiphase RSM
based on the data of point-particle DNS. The RSM framework has been used to simulate
particle-laden vertical channel flows (Capecelatro et al. 2016a; Baker et al. 2020). Riella
et al. (2019) proposed a pressure-velocity model for both particle and fluid phases that
can account for the strong near-wall inhomogeneity of particle-laden channel flows. From
experimental data on bubbly pipe flows, Colombo & Fairweather (2015) developed a RSM
based on the Speziale—Sarkar—Gatski (SSG) model (Speziale, Sarkar & Gatski 1991) for
bubble-induced turbulence.

In the past decades, interface-resolved (or particle-resolved) DNS (IR-DNS or PR-DNS)
has been extensively employed to investigate the mechanisms in turbulence—particle
interactions and the development of the multiphase turbulence model (Balachandar &
Eaton 2010; Tenneti & Subramaniam 2014; Maxey 2017). Ma et al. (2017) for the first time
used the IR-DNS data of bubble-induced turbulence in a vertical channel flow to develop
the interfacial term models. In particular, a correlation of C,3 = 0.3Cp is proposed, as
shown in table 1, at the bubble volume fraction ranging from 1.07 % to 2.14 % and the
bubble Reynolds number above 200. More recently, Ma et al. (2020b) extended this work
in the framework of a second-moment closure for bubble-induced turbulence from DNS
of dispersed bubbly flow in a vertical channel, and modelling of the pressure—strain term
and the interfacial term in Reynolds stress equations was taken into consideration.

To our knowledge, the works on IR-DNS-based multiphase turbulence modelling are
limited to bubble-induced turbulence at relatively low bubble volume fractions (Ma et al.
2017, 2020b). The IR-DNS data so far have not been used to develop a dissipation rate
model for particle-laden flows. Thus, we aim to develop new models or correlations for the
interfacial terms in the dissipation rate equation as well as in the Reynolds stress equations,
based on interface-resolved simulations of particle sedimentation in a periodic domain at
a density ratio ranging from 0.01 to 1000, a particle concentration ranging from 2.3 % to
30.2 % and a particle Reynolds number below 250. The correlations for the mean drag and
the pseudo-TKE are also presented, which are required by the modelling of the interfacial
term in the dissipation rate equation. In addition, the new models are employed in the
RSM-based Reynolds-averaged Navier—Stokes (RANS) simulations of fully developed
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vertical particle-laden channel flows. It is found that the model predictions are in good
agreement with the DNS results for both turbulence enhancement and attenuation regimes.

The rest of the paper is organized as follows. In § 2 we address the basic equations of
the RSM for the particle-laden flows. In § 3 we present the numerical method and the
set-up for our interface-resolved simulations. Next, in § 4 the coefficients of the interfacial
terms in the dissipation rate and Reynolds stress equations are determined on the basis
of our DNS data. Then, in §5 the proposed model is applied to RSM-based RANS
simulations of vertical particle-laden channel flows, and the results are compared with
the interface-resolved simulations. Concluding remarks are summarized in § 6.

2. Reynolds stress model

The averaged continuity and momentum equations of the Euler—Euler two-fluid approach
for particle-laden flows (Anderson & Jackson 1967; Crowe et al. 2011) are

KOnP) 4 Gupatn)) = 0. @.1)
YGOD 5 Gyortugh ) = Vo) + - @rleg)
V- @yl + drorg fs opnds, 22)
HOLBD, 15 ooyl ) = ~6,¥ o) +V - (9s{0)
— V- (¢sps(Uitdl)) + dspsg + ‘l/ /51 oy nds. (2.3)

The subscript m in (2.1) denotes the phase (f denoting fluid phase and s solid phase). Here
o, p, u, u', T and o represent the density, pressure, velocity, fluctuating velocity, viscous
stress tensor and total stress tensor, respectively; (-) represents phase averaging; ¢ and
¢s denote the mean volume fraction of fluid and solid phases, with ¢y + ¢5 = 1.0. The
last term in (2.2) or (2.3) is the volume-averaged interphase hydrodynamic force, which
may include the steady drag force, lift force and history force (Saffman 1956; Wen &
Yu 1966; Antal, Lahey Jr & Flaherty 1991; Gidaspow 1994; Burns et al. 2004; Tenneti &
Subramaniam 2014). Here we are only concerned with the steady mean drag force. In (2.2)
the term —¢fpf(u}u}) is the fluid Reynolds stress tensor. In (2.3), o is the solid stress,
including the solid pressure and shear stress, which are generally modelled by the kinetic
theory of granular flows (Fox 2014).

The transport equations for the fluid-phase Reynolds stresses can be written as (Kataoka,
Besnard & Serizawa 1992; Crowe et al. 2011; Peng, Ayala & Wang 2019)

A (Broplua)) (b pr (ux) (uad})) L ) 1)
ot + ™ = - ¢f/0f((uiuk)a—xk + (ujuk> ™ )
Py
+ Dy + P — 5 + 1. (2.4)

Here and in the following, the subscript ‘f” for the fluid velocity, pressure and stress
is omitted for convenience, unless otherwise specified. The shear-induced production
of Reynolds stresses, P, can be determined without modelling. Here Dy, ¢, @;; and
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I represent the diffusion, pressure—strain term, dissipation term and interfacial term,
respectively, and they need to be modelled for the second-moment closure. The diffusion
term can be expressed as

@rpr i) [a(qsf(u;p/)) . 8(¢f’<”}l7/>)j|

0Xk

il'_

20X ox;

ArluiTy))  0(druiTy))
+ fAYU; jk + f J ik , (25)
0xXk 0Xk
and can be modelled with the gradient diffusion hypothesis as (Shir 1973)
9 a(u;u]’.)
Djj = I drpr(v + cgvp) o | (2.6)

where the eddy viscosity v; and an empirical coefficient ¢, are introduced.
The dissipation term ¢;; can be expressed as

, ou; , 0
gjj = —¢f 'L'l-ka—xk + Tjka—xk . (27)

The isotropic model of &;; = —¢r pf%&je is widely used for high-Re turbulent flows, where
¢ is the average fluid-phase TKE dissipation rate. For low-Re wall-bounded flows, an
anisotropic model proposed by Hanjali¢ & Launder (1976) is widely used, i.e.

() 2
&ij = ¢ror | fs . +§(1 — 8| e (2.8)

where f; is normally chosen as a constant or a function of the turbulent Reynolds number
Re; = k*/ev. For f; = 0, &jj is isotropic, whereas for f; = 1, ¢;; is fully anisotropic (Rotta
1951). For the particle-free channel flow, the DNS data of Kim, Moin & Moser (1987) and
Mansour, Kim & Moin (1988) revealed that the dissipation tensor is strongly anisotropic
with f; & 1 close to the wall.

The pressure—strain term @;; has the following form:

PPN WL S 2.9)
v f 0x; ! ax; | '

The pressure—strain term plays a significant role in predicting the Reynolds stress
anisotropy, and is most difficult for accurate modelling. It is generally modelled by the
SSG model (Hanjali¢ & Launder 2011)

Pjj = Djj | + Pjj2 + Pij3 + Pijws (2.10)

where @1, @2, ®;3 and Pj;,, represent the return to isotropy of turbulence (the
slow term), isotropization of strain production (the rapid term), isotropization of force
production and the wall blocking (Hanjali¢ & Launder 2011), respectively. The terms ®; 1
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and @;; > are modelled as (Lumley 1979; Speziale er al. 1991)
@1 = —drprCreaj + ¢rprCle(aiar — 3A28;), (2.11)

Dijr = —¢pprCa(Pyj — 3PiSyj) — drprCa(Ejj — FP8y)

— ¢y pr(0.033 + 0.65A3 *)kS;; — ¢y prCraiPy. (2.12)
The isotropization of force production @;; 3 can be modelled as (Naot 1970)
@3 = —C3(ly — 3I5y). (2.13)
In the above equations, a;; is the fluid Reynolds stress anisotropy tensor
1 Lo 2
aj = z ((uiuj) — gk(sij) , (2.14)

and Ay = aj;a;; is the second invariant of a;;. The coefficients Cy, C/l, (2, C3, C4 and C}
are model coefficients; Py represents the TKE production rate; E;; and S;; are given as

/7 a(”k) /i 8(”]()
Ej=— |:(uiuk>8—xj + (ujuk>8—xi] ) (2.15)
1) | dw)
Sij = > ( o, + o ) . (2.16)

The Reynolds stress interfacial term /;; can be expressed as
1
Ij = -7 |:/S, nu;(—p'Sjic + r]-/k) ds + /SI nku;(—p’(Sik + 74) dsi| , (2.17)
where S is the interphase surface inside the averaging volume. Neglecting the effect of the
particle rotation, Crowe et al. (2011) derived the following form for the interfacial term:

N, N,
1 <& 1 <
ly = ij«u» — gk )Fij + ij () — tprj)Fr.i- (2.18)

Here uyy ; denotes the ith velocity component of the kth particle, N, denotes the number
of particles in the volume V and Fy ; represents the drag force on the kth particle. Xia et al.
(2022a) showed that, for the TKE interfacial term at the steady state, the term related to
the particle rotation and the correlation between the particle fluctuating velocity and the
drag force are negligibly small. Thus, the interfacial term (2.18) can reduce to

lij = Fpiuyj + Fpjuyi, (2.19)

where Fp is the volume-averaged drag and u, represents the mean interphase slip velocity.
On the other hand, it is conventional to establish the model of the Reynolds stress
interfacial term with the TKE interfacial term I, (Ma et al. 2020b)

lij = bjjl, (2.20)

where bj; is a coefficient matrix. The Troshko—Hassan model (Troshko & Hassan 2001) is
one of widely used models for the TKE interfacial term,

Ii=Fp-u,. (2.21)

It was suggested to multiply (2.21) by a correction coefficient Cy,, i.e. Iy = C,Fp - u,.
In the ANSYS FLUENT software, a value of 0.75 is recommended for C, (Ansys, Inc
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2013). Here C,, was determined as a function of TKE or particle Reynolds number from
DNS of bubbly flows (Santarelli, Roussel & Frohlich 2016; Ma et al. 2017) and as a
function of the particle Reynolds number, the particle size, the density ratio and the
particle volume fraction, based on the DNS data on the upward turbulent channel flow (Yu
et al. 2021). However, we have recently demonstrated that C,=1 is a better choice for the
particle-laden flows at a statistically steady state (Xia et al. 2022a), being consistent with
the observation of Mehrabadi et al. (2015). For the particle sedimentation in a periodic
domain or in a vertical channel flow considered in the present study, the mean drag and
the mean slip velocity are only non-zero in the vertical direction, thus, /1] = 2I; and the
other components are zero. Accordingly, the only non-zero component of b;; is by = 2.0.
The transport equation of the fluid-phase TKE reads

A(Prork) | 3(@rorlupk) L0 u)
o oy Prpr () 3%,
Py
0 ok
+ | bror(v +esv) o— | — drpre +ik (2.22)
8Xk 8Xk ———
&k
Dy

where Py and Dy represent the production and the diffusion of TKE, and ¢ is the
dissipation term of TKE.

The transport equation for the dissipation rate ¢ is given as
;g 0{ui)

d(drpre) . d(drpr(uj)e) €

o1 ox  PrArCenluin)
PS
L2 (¢fpf (v + E) 3—8) —Cotyry S+ (2.23)
0x; oz ) 0x; Tk
D, Ee

Here P,, D, and ¢, represent the production, diffusion and dissipation of TKE dissipation
rate, respectively; I is the interfacial term in the ¢ equation; C¢1, Cs> and o, are model
constants.

3. Numerical method and DNS setup
3.1. Direct forcing/fictitious domain method

The IR-DNS approach used in this work is the direct forcing/fictitious domain (DF/FD)
method (Yu & Shao 2007; Yu et al. 2016). The key idea of this method is to fill the
interior domain of solid particles with fictitious fluids, and introduce a pseudo-body
force (i.e. distributed Lagrange multiplier Glowinski et al. 1999) over the solid region
to enforce the rigid-body constraints. For simplicity of description, we consider only one
particle in the following formulae. Let P(¢) represent the solid domain and 2 represents
the entire domain including both the interior and exterior of the solid body. The particle
density, volume and moment of inertia tensor, translational velocity and angular velocity
are denoted by p;, V), J, U and w), respectively. The viscosity and density of the fluid are
w and pf.
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We introduce the following characteristic scales for the non-dimensionalization: L.
for length, U, for velocity, L./U, for time, ,ofUcz/Lc for the body force. Then the
dimensionless FD formulation for incompressible fluid can be written as

o vu= Y ot aing (3.1
_ 170 Uu— ——— — mn , .
at Re P
u=U+w, x rinP(1), 3.2)
V.u=0in £, 3.3)
dU g
- OV — —Fr2 ) =— [ Adx, 34
(or ),,(dt rg) f,, (3.4)
dJ* .
(pr—l)wz—/rx/ldx, (3.5)
dr P

where u, p and A, r represent the fluid velocity, pressure, pseudo-body force and position
vector with respect to the mass centre of the particle, respectively. Here p, is the
particle—fluid density ratio defined by p, = p;/pr, Re the Reynolds number defined by
Re = prUcL./jt and Fr is the Froude number defined by Fr = gL/ Uf, with g being the
gravitational acceleration; V[’f and J* are the dimensionless particle volume and moment
of inertia tensor, defined by V; =V, /L and J* = J/psL>.

A fractional-step temporal scheme is used to decouple the system (3.1)—(3.5) into two
subproblems.

(1) The fluid subproblem for #* and p is given by

ut—u" 1 Vu* v 1[3( Vay' — (- Vuy™] 4 1 V2" o
- = =—-Vp—-[B(u-Vu)"'— (u-Vu —— ,
At 2 Re P=3 2 Re 6

V.u=0. (3.7)

A finite-difference-based projection method on a homogeneous half-staggered grid
is used. All spatial derivatives are discretized with a second-order central difference
scheme.

(2) The particle subproblem for U"+!, ot A and 4"+ is given by

Un—H U" g u*
A = -HVvi|— —Fr= / — = A" dx, 3.8
J*. wn+1 T ol u*
p _ P n * n n

prT_(pr—l)[ A7 —wpx(J-wp)}+ﬁrx(E—A>m.

(3.9
The pseudo-body forces A are then updated from
Un+1 4 wn+1 X r—u*
A = L e 3.10
A + (3.10)
Finally, the fluid velocities #"+! at the Eulerian nodes are corrected as

u™h = w4+ A - A7), (3.11)

In the above manipulations, a discrete § function in the form of a trilinear function is
employed to transfer quantities between the Eulerian and Lagrangian nodes.
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3.2. Collision model

We implement a soft-sphere collision model with lubrication correction to solve the
short-ranged inter-particle interaction. A discrete element model based on a nonlinear
soft-sphere approach is coupled to our DF/FD solver. The total contact force acting on
particle i by another particle j is described by

2
Fijn = (—ka8)* = nauy - mm, (3.12)
Fij,t = (—kt& — T]tllij’[ . t)t, (313)
where Fjj ,, 8,, ky, and 1, are the normal contact force, overlapping distance, stiffness
coefficient and damping coefficient, respectively, and F'j; ;, &, k;, 0, are the corresponding
parameters in the tangential direction. Also, n is the unit normal vector pointing from the
centre of particle 7 to that of particle j, and ¢ is the unit tangential vector. If u;; = u; — u;

denotes the relative velocity between particles then the relative normal velocity can be
written as u;; , = (u;; - n)n, and the slip velocity at the contact point is

Ujjr = Wjj — Wjj n + (@jw; X n + ajw; X n), (3.14)

where a denotes the radius of the particle and w is the rotational velocity of the particle.
The values of &, and k; are given using the Hertzian contact theory and Mindlin’s theory
(Crowe et al. 2011)

—1
4f1-02 1-0? a2
=T ) (SR (3.15)
3 E; E; a;a;
. —o\" /g N\ —1/2
kt -3 (2 — Oj + 2 G]) (at +a]> 6’,11/2, (316)
G Gj aa;j

where E and o are the Young’s modulus and Poisson ratio, respectively, and G =
E/2(1 + o) is the particle shear modulus. Damping coefficients are given as (Barnocky &
Davis 1988; Tsuji, Tanaka & Ishida 1992)

M = bu/mpkn8l/4, 0y = y/mkss)’, (3.17a.b)

where m, = m;m;/(m; + m;) is the effective mass, ¢, is a constant related to the dry
coefficient of restitution e; as shown in Tsuji et al. (1992). If |F,| > f|F,| then the
tangential force is given by the Coulomb-type friction law

Fi = —f|F.lt, (3.18)

where f is the friction coefficient. The time step for the collision is set to be one tenth of
that for the flow solution (i.e. Az, = At/10).

Since the DF/FD method is a fixed mesh method, the correction of the lubrication force
is needed when two spheres are very close to each other. The lubrication force corrections
have the form

Fij jup = —6mprau; n[A(€) — A(€ar)], (3.19)

where €,; = Ax/a is the threshold gap below which the lubrication correction is activated,
where a represents the radius of the particle and Ax is the grid size, and A(¢) is a function
of the normalized gap distance € = ¢, /a (Jeffrey 1982), so

1 9 3
Ale) = — — —1 — —€l 0.673 + O(e). 3.20
(€) = 7= = ;g€ — pzeln@) + + 0(e) (3.20)

Note that the lubrication correction is kept constant for € < €1 to account for the
lubrication force saturation due to the effect of the surface roughness. In addition, the
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(b) () (d)

2L 2L

Figure 1. Sketches of the particle sedimentation in a fully periodic domain for three different particle volume
fractions: (a) ¢ = 2.3%, (b) ¢ =5.0%, (c) p = 14.2%, (d) ¢ = 30.2 %.

¢ 23% 5.0% 14.2 % 30.2 %
Ny 11790 241192 68 /540 144/ 1152
Ga 24.2-194.0 24.2-218.2 24.2-242.5 24.2-315.2

Table 2. Simulation settings for particles settling in a fully periodic domain at p, = 2.0. Here N, is the total
particle number and ¢ is the mean solid volume fraction. Because we have two computational domain sizes,
the particle numbers before the ‘/* correspond to L/d, = 5 while the numbers after the ‘/° correspond to
L/d, = 10.

lubrication force correction is turned off for € < —e> when two particles overlap and the
squeezing fluid film is damaged. Here €; and €, are set to be 0.001 in our simulations.

3.3. Simulation set-up

In the present study we focus on modelling of the interfacial terms for the case where
the mean slip velocity exists. Thus, we use the DNS data on the particle sedimentation
to establish the correlations for the interfacial terms. As shown in figure 1, the particles
are freely settling in a periodic cuboid, with the size of Ly x Ly x L, = 2L x L x L. The
domain is discretized using a uniform Cartesian grid, and the spatial resolution with
respect to the particle size is dj,/Ax = 25.6. The mean velocity of the suspension in the
direction of gravity (i.e. x-direction) is kept zero by instantaneously adjusting the pressure
gradient.

The parameter settings for our simulations are listed in tables 2 and 3. For convenience,
we use ¢ to denote the particle (or solid) volume fraction. The particle number N,, is given

by
66 (L\°
- _<_) . (3.21)

T \dp

We select four solid volume fractions ¢ = 2.3 %, 5.0 %, 14.2 % and 30.2 %, and two
domain sizes: L/d, =35 and 10 for the liquid—solid case of p, = 2.0, as shown in
table 2. We also perform some simulations for p, = 0.01 and 1000, which represent
the bubble-liquid and gas—solid systems, respectively, as shown in table 3. Numerical
instability was encountered for p, = 0.001 at high particle volume fractions, hence, p, =
0.001 is not considered.
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¢ Np Pr Ga

14.2 % 540 0.01, 1000 24.2,72.7,169.7, 242.5
30.2 % 1152 0.01, 1000 24.2,72.7,169.7, 315.2

Table 3. Simulation settings used for particles settling in a fully periodic domain of L/d,, = 10 for various
density ratios.

We take the sphere diameter d), as the characteristic length, and the characteristic

velocity u, is selected as
4d,
Ue = T(pr - Dg. (3.22)

The Galileo number is widely used to measure the particle settling effect, which is

defined as
J&(or — g
e (3.23)

1%

Ga =

The mean relative (or slip) velocity u, is defined as the difference between the average
solid and fluid-phase velocities directed along the x axis as

ur = (ur) — (up,1). (3.24)

The particle Reynolds number is defined as Re, = u,d),/v.

We set E/(pru?) =2 x 10*, o = 0.33 and f = 0.3 for the collision model. The
dimensionless time step is At = 0.005 and the statistics are obtained from averaging the
data for a duration of 1000 non-dimensional time unit (d,/u.) after the statistically steady
state is reached. The statistics for the solid-phase velocity and volume fraction are obtained
from the data in the solid domain (i.e. inside the particle boundary). Note that we have
examined the effects of the box size and the mesh resolution on the results in Appendix A,
and ensured that our results are not affected significantly by the box size and the mesh
resolution.

4. Modelling of interfacial terms
4.1. Modelling of dissipation interfacial term

Using the spatial averaging theorem, Schwarzkopf et al. (2009) derived the dissipation rate
transport equation of the fluid phase, in which the interfacial terms due to the presence of
particles have the form (see also Crowe et al. (2011) for the derivation)

2 / / / / 2 / /

0 au; ou; 2 au’; 0 0 [du; du;
Isz—ﬂ—/ o ds + 2 —’inidS—ﬂ/ A e e T
Vo oxg Js, 0xj 0x; V. Js, 0x; 0x; Vo Js, 0xx [ 0x; 0x;

4.1)

The first and second terms on the right-hand side of (4.1) are not important, as
demonstrated by Schwarzkopf et al. (2009). For the Stokes flow in the dilute limit, the
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last term of (4.1) can be simplified as (Schwarzkopf et al. 2009; Crowe et al. 2011)

2 / / 2 2
0 ou; ou; : i) — i
V. Js, 0xx \ 0x;j 0x; Vv P d, d;

The constant ‘24’ can be replaced by the so-called dissipation interfacial term coefficient
C¢3 to account for non-Stokesian effects, resulting in

V
I. = Ce3— 1. (4.3)
dP

Accurate computation of the interfacial term via the integration of (4.2) on the
particle surface is difficult for the non-body-fitted approach. Therefore, we quantify it by
considering the dissipation rate balance equation (Schwarzkopf et al. 2009). For the fully
developed homogeneous system with dk/dt = 0, de/df = 0 and 9(-)/dx; = 0O, the k and ¢
equations reduce to

I = ¢rpre, 4.4)
&2
I, = ngqbfp_f?. 4.5)
From (4.3)-(4.5), we have
Cy &b
S _Tnf 4.6)
Ceo v k
From (2.21) and (4.4), one can obtain
F
g = 21 @.7)
bror
Substituting (4.7) into (4.6) yields
Ca db 1 d,F.
=S P Fpu, =Re, L, 4.8)
Co v rork (1= §) oy (k/u2)u

which indicates that C.3/Cy is related to the relative TKE &/ u% and the mean drag force
Fp. Hence, to model the interfacial term in the dissipation rate equation, the correlations
for the mean drag and the relative TKE are required.

Based on interface-resolved simulations of the flow past fixed random assemblies
of monodisperse spheres (0.1 < ¢ < 0.5, 0.01 < Re,,, < 300, where Re,, = (1 — ¢)Rep),
Mehrabadi et al. (2015) proposed the following correlation for the particle-induced relative
TKE:

k 3 172
o ¢+ 1.25¢ (1 — ¢)” exp(—¢Re,, ). 4.9)

:
This was shown to be applicable to the dynamic sedimentation system. In the present
work we attempt to quantify the relative TKE for the dynamic sedimentation system at
different density ratios. Figure 2(a) shows the variation of k/u? with the particle Reynolds
number for different solid volume fractions at a density ratio of 2. k/u? for two different
box sizes of L/d, = 10 and 5, which shows that k/uf slightly increases with increasing
domain size. Figure 2(b) shows the effects of the density ratio on k/u? for ¢ = 14.2%
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Figure 2. Relative TKE k/ uf as a function of Re,, for (a) p, = 2.0 and different domain sizes, and (b) o, =
0.01 and 1000 at L/d, = 10. The symbols represent the IR-DNS data, while the lines denote the corresponding
correlations (4.10). In the legend of (a), ‘small’ represents the smaller domain of L/d, = 5, while ‘large’
represents the larger domain of L/d,, = 10.

and 30.2 %. From figure 2(a), k/ uf decreases rapidly with increasing particle Reynolds
number at relatively small Re,,, particularly for dense suspensions, consistent with previous
simulations of Mehrabadi et al. (2015). For dilute suspensions, we observe a slight increase
in k/u? with increasing Re, at Re, > 100. From figure 2(b), k/u? is higher for a lower
density ratio. Using the present DNS data (0.023 < ¢ < 0.302, Re, < 250) and inspired
by the correlation function form of Mehrabadi et al. (2015), a correlation for the relative
TKE is obtained as

k
= 0.92p, %2497 1 0.05¢% (1 — ¢)(Re,/200)"!

u_% r
+25¢%(pr + 0.5 751 — ¢)’ exp(—¢'/2(1 — ¢)/?Re,/?). (4.10)

Figure 2 shows that the above correlation agrees well with our DNS data. Figure 3
compares the present correlation of the relative TKE for p, = 1000 to that of Mehrabadi
et al. (2015) obtained from the simulations of fixed spheres. Both correlations are in fairly
good agreement with each other. One discrepancy is that the relative TKE of Mehrabadi
et al. (2015) decreases monotonically with increasing Re),, whereas ours slightly increases
with increasing Re), for Re;, > 100. Mehrabadi et al. (2015) observed that the relative TKE
for freely moving particles was slightly greater than that for fixed particles at relatively
high Re,. Our computational domain size is larger than that of Mehrabadi e al. (2015),
and it has been observed that the relative TKE increases with increasing domain size.
Therefore, the differences in the domain size and particle mobility may be responsible for
the above discrepancy.

We now intend to establish the correlation for the mean drag from our DNS data, which
is also required for modelling the dissipation interfacial term (4.8). We take the following
form for the drag correlation

Fp = Buy, @.11)

where $ is the momentum transfer coefficient.
For ¢ < 0.1, we adopt the correlation of 8 obtained in our previous work (Xia et al.
2022b),

3

1 — r
g = 4Cdpf¢( Pu

a, (I=¢), (4.12)
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Figure 3. Comparison of relative TKE k/ uf obtained from the present simulations and from the correlation of
Mehrabadi er al. (2015). Our IR-DNS data for p, = 1000, denoted by the symbols, and our correlation (4.10),
denoted by the lines, are both shown.

where Cjy is the modified drag coefficient

24

_ _ 0.687
= - q))Rep(l +0.15((1 — ¢)Rep) ), (4.13)

Cu

and n is the exponent of the correction function in the form
n = (3.88 — 0.38In(Rey)) In(¢p 1) p, 038, (4.14)

For ¢ > 0.1, based on our DNS data and the modification of the drag correlation of
Tang et al. (2015) by considering the effects of particle mobility and density ratio, we
obtain the following drag correlation:

> (1 — o1 +1.5J¢)  18u¢(l — d)Rey [
=180 18 0.11¢ (1
B2 20— ) + pa + &2 o(1+¢)

0.00456 0.0644

— =+ (01691 — — R 0-343]
(1—¢>4+< ( ¢)+(1—¢>4> "

2
2. 76ReD8 50057 (4.15)

E(1—¢)

We note that in order to avoid the discontinuity of B at the critical solid volume
fraction (e.g. 0.1 here), Lu et al. (2003) proposed a continuous stitching function as
B =«kp1+ (1 —«)Ba, with x = arctan[150 x 1.75(0.1 — ¢)]/7 + 0.5 being an inverse
tangent function.

The mean drag forces Fp obtained from IR-DNS normalized with 37w (1 — ¢)d,u,/ dg
at different Re, and ¢ are compared with the present drag correlations in figure 4. The
correlations agree well with our DNS data for all cases considered, as shown in figure 4.
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Figure 4. Mean drag forces Fp, normalized with 3 (1l — ¢)dpu,/d3, as a function of Re), for different ¢,
and (a) pr = 2.0, (b) pr = 0.01-1000 and L/d,, = 10. The symbols denote the IR-DNS data, while the lines
present the corresponding drag correlation (4.12) or (4.15). In the legend of (a) , ‘small’ represents the smaller
domain of L/d, = 5, while ‘large’ represents the larger domain of L/d, = 10.
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Figure 5. Coefficient ratio C¢3/C;2 as a function of Re,, at different solid volume fractions and (a) p, = 2.0,
(b) pr =0.01 and 1000, and L/d, = 10. The symbols denote the IR-DNS data, while the lines denote the
correlation (4.16).

With the established correlations for the relative TKE (4.10) and the mean drag
coefficient (4.11), the ratio C¢3/C,2 can be modelled as

Cos BRe,  d, @16

Co pr(1—¢)(k/ud) u,

The comparison of this correlation with the IR-DNS data is shown in figure 5. From
figure 5(a), Cg3/Cs7 increases with increasing particle Reynolds number or solid volume
fraction, when the other parameters are fixed. In addition, the effect of the computational
domain size becomes more pronounced at a higher solid volume fraction, and C¢3/Ce2
is larger for a smaller domain size. Figure 5(b) shows that C.3/Cs2 is much higher
for p, = 1000 than those for p, =2 and 0.01, which indicates that the dissipation rate
interfacial term is higher for particles with a larger density at the same particle Reynolds

number and solid volume fraction.

In the dilute limit, C¢3 can be simplified as
3

Ces = ZCSZJ;M%CDR%. 4.17)
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Reference Carrier  Disperse Rey, Rep dy/2H Or b0

Santarelli & Water Bubbles 5263 236-475 0.052-0.076  0.01  0.0029-0.0214
Frohlich (2015),

Santarelli er al.

(2016)

Uhlmann (2008) Water Solid particles 5400 136 0.05 2.21 or 10 0.0042

Yu et al. (2021) Water Solid particles 5746, 12000 50-225 0.05,0.1  2.0-100 0.0236

Table 4. Summary of the previous works on interface-resolved simulations of vertical turbulent
particle-laden channel flows.

Therefore, the interfacial term of dissipation rate /, can be expressed as

3 ¢ v 3 ] Ix
I, = -Coo——=CpRep,— 1y = —Cep——Cp—, 4.18
& 4 82k/u% D epdl%k 4 82k/u% D‘C ( )
where T = d), /u,. Our model in the dilute limit (4.18) has a form similar to that of Troshko
& Hassan (2001) (I, = 1.35Cpl/t) and Ma et al. (2017)(I = 0.3Cpl/t), as shown in
table 1.

From (4.8), we have

C.: d,F
[(1 —¢)(k/uf)c—2]/[;;u§] = Re,. (4.19)

We collect the IR-DNS data on vertical channel flows laden with bubbles or heavy particles
in the literature (Uhlmann 2008; Santarelli & Frohlich 2015; Santarelli er al. 2016; Yu
et al. 2021), as summarized in table 4, and compare them with our model in figure 6. It
is noted that the values of C¢3/Cy in figure 6 are calculated directly from (4.6), and Fp
and k/ uf are computed with our correlations (4.10) and (4.11). All data are collected in the
sedimentation-dominated regime of v/k/u, < 1 at the channel centre. Our model agrees
very well with the DNS data of Yu et al. (2021) for particle Reynolds numbers below
200, and is in fairly good agreement with those of Uhlmann (2008), Santarelli & Frohlich
(2015) and Santarelli et al. (2016).

4.2. Modelling of the interfacial term in Reynolds stress equation

The presence of particles gives rise to strong anisotropy for the Reynolds stress transport.
For the homogeneous sedimentation system at steady state, the Reynolds stress equations
(2.4) reduce to

Considerable efforts have been devoted to the modelling of the dissipation rate tensor &;;,
pressure—strain term &;; and interfacial term /;;.

Poelma, Westerweel & Ooms (2007) have demonstrated experimentally that relative
motion between the phases induced by gravity forces can lead to significant anisotropy
in the fluid-phase dissipation tensor &;. Using the present DNS data, we determine the
coefficient f; in (2.8), and the results for different Re, and ¢ at p, = 2.0 are plotted
in figure 7(a). Baker et al. (2020) adopted f; = 0.9. Figure 7(a) shows that f; from our
DNS data is insensitive to the variation of Re, and ¢, and f; = 0.9 is a good choice.
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Figure 6. Previous IR-DNS data on C¢3/Cy at the channel centre for vertical channel flows, as compared
with the model (4.19).
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Figure 7. Plots of (a) f; and (b) C3 as a function of Re,, for different solid volume fractions and p, = 2. The
symbols denote our IR-DNS data, and the solid lines represent the model (4.22).

It should be noted that the results for p, = 0.01 and 1000 are similar to those for p, = 2.0
in figure 7(a), but are not shown here for clarity.
For the homogeneous sedimentation system, the pressure—strain term @;; reduces to

@ = —¢yprCreaj + dpprCle(aiay — $A28;) — C3lj — 8ilkk)- (4.21)

With the standard coefficients C; = 1.7 and C} = 1.05 for the SSG model (Hanjali¢ &
Launder 2011), the only unknown coefficient C3 can be obtained using the DNS data from
(4.21). The results are shown in figure 7(b). Obviously, C3 increases with increasing ¢.
For dilute flows (¢ = 2.3 %) at Re, < 150, C3 is almost zero. For p, = 2, C3 is a function
of particle volume fraction and particle Reynolds number in the form

C3(¢.Rep) = ¢ + aRep + bRe, + c, (4.22)

in which a =3.723 x 1074, b = 2.819 x 107° and ¢ = —0.010. The correlation of C3

(4.22) is compared with the DNS data in figure 7(b), and good agreement can be observed.
We note that C3 is not sensitive to the variation of the density ratio from our DNS (not
shown here), thus, the correlation (4.22) is applicable to other density ratios.
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Figure 8. Reynolds stress anisotropy component ay; for different Re, and ¢ at (@) por = 2.0 and (b) pr =
0.01-1000. The symbols denote our IR-DNS data, while the lines represent the model proposed by Mehrabadi
et al. (2015) from DNS of fixed particles.

The non-monotonic trend of C3 with Re, may be related to the particle-induced fluid
Reynolds stress anisotropy a;; defined by (2.14), since Cj3 is determined from (4.21) with
the DNS data of a;;, ®;; and /;;. Our results on the Reynolds stress anisotropy component in
the vertical direction aq; are shown in figure 8. From figure 8, a1 increases with increasing
Re, at relatively low Rep, and then decreases with increasing Re), above a critical Re),
being opposite to the dependence of C3 on Re),, as shown in figure 7(b). From figure 8, ay;
decreases with increasing ¢, which indicates that the anisotropy of the velocity fluctuation
decreases with increasing ¢, since ai; + a2 + a3z = 2 and ayy = azz. Mehrabadi et al.
(2015) argued that the magnitude of anisotropy is determined by the ratio of the length
of elongated structures to the inter-particle spacing (Mehrabadi et al. 2015). In addition,
figure 8(b) shows that ay; increases with increasing density ratio, particularly for high Re),,
and ay; for p, = 1000 is in good agreement with the algebraic stress model proposed by
Mehrabadi et al. (2015) obtained from the DNS data for fixed spheres, which provides
evidence for good accuracy of our DNS data.

The force term @;; 3 with the coefficient of C3 in the pressure—strain model is caused by
the presence of particles, and this term can be combined with the interfacial term /;;. Ma
et al. (2020b) defined an effective particle-induced interfacial term I;; = &3+ Ijj, ie.

I} = byl — C3(Ij — 385I) = [(1 — C3)bj + 3C38;1Ix. (4.23)
Therefore, we have IlT’J‘. = b;“jlk, with

bZ =(1- C3)bl] + %CQ,(SU (4.24)

As discussed in § 2, b1; = 2.0 and by = b33 = 0, if the mean slip velocity exists only
in one direction. Figure 9 shows b}, as a function of Re, for different ¢ at p, = 2.
Clearly, b’fl decreases with increasing ¢, as ap; in figure 8, which indicates a trend
towards stronger isotropy at a higher ¢. For ¢ = 2.3 % and Re, < 150, our b7}, is close
to 2.0, which is consistent with the correlation of Ma et al. (2020b) from DNS of bubbly
flows at relatively low particle volume fractions and high particle Reynolds numbers:
b, = min(0.67 + 0.67 exp(370Re, %), 2) (Ma et al. 2020b).

5. Reynolds-averaged simulation of turbulent channel flows

It remains unclear whether the existing multiphase turbulence models can correctly
predict the turbulence enhancement or attenuation induced by particles. In this section
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Figure 9. Modified interfacial term b}, as a function of Re,, for different ¢ at p, = 2. The solid line
represents the correlation of Ma et al. (2020b).

Case Or dy/2H b0 Ga Re, Rep, Flow direction Source

1 2.0 0.05 2.36 % 22.9 9.5 5746 Upward Yu

2 2.0 0.05 2.36 % 62.2 49.6 5746 Upward Yu

3 2.0 0.05 2.36 % 112.0 113.5 5746 Upward Yu

4 2.0 0.05 2.36 % 103.9 102.1 12000 Upward Yu

5 2.0 0.05 2.36 % 49.8 38.3 5746 Downward Xia

6 100 0.05 2.36 % 112.0 90.6 5746 Downward Xia

7 2.21 0.025 0.42 % 115.5 130.5 5400 Upward Uhlmann

8 0.001 0.052 2.14 % 195.4 235.5 5263 Upward Santarelli
9 2.0 0.05 14.2 % 50.0 22.9 5746 Upward Present work

Table 5. Key parameters for the selected particle-laden flows in a vertical channel (Uhlmann 2008; Santarelli
et al. 2016; Xia et al. 2021; Yu et al. 2021). Here ¢y is the mean solid volume fraction in the entire channel,
Rep = up(2H) /v is the bulk Reynolds number, Re), is the particle Reynolds number based on the mean
interphase slip velocity at the channel centre.

the particle-laden flows in a vertical channel at relatively low particle volume fractions are
simulated using the RSM with the interfacial term correlations developed in the present
study. The results are compared with the IR-DNS data of Yu et al. (2021), Xia et al. (2021),
Uhlmann (2008) and Santarelli et al. (2016) (see table 5 for the parameter settings), in
order to validate our new models. In addition, a new IR-DNS simulation with a particle
volume fraction of 14.2 % (case 9 in table 5) is performed in the present work to validate
the accuracy of our model for relatively large volume fractions.

5.1. Near-wall RSM for particle-laden turbulent channel flows
Most turbulence models were developed for high Reynolds numbers, and they are not
suitable for low-Re wall-bounded channel flows where the viscous effects permeate the
whole region. The Reynolds numbers for all IR-DNS data available are low, hence, the
near-wall Reynolds stress model (So et al. 1996) for the single-phase turbulent flows is
employed. In the present study we assume that the presence of particles do not affect
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Figure 10. Comparison between the dissipation rate tensor ¢; obtained fully from the DNS simulations and
from the model (2.8) with f; = 0.9 and the DNS data of the dissipation rate ¢ for (a) case 1 (Re, = 9.5) and

(b) case 3 (Re, = 113.5). All values are normalized with pfug /H.

the model parameters existing in the single-phase case in (2.4) and (2.23), and we only
supplement with the particle-induced terms.

For the dissipation term, the model (2.8) is employed with f; = 0.9, as discussed earlier
in association with figure 7(a). The components of the dissipation term ¢; predicted by
the model (2.8) using the DNS data of the dissipation rate ¢ and Reynolds stresses are
compared with the DNS data of ¢; for the typical cases (cases 1 and 3) in figure 10.
Remarkably good agreement can be seen, which indicates that the anisotropic model (2.8)
with f; = 0.9 is suitable for the particle-laden channel flow.

From (2.10)—(2.13), the pressure—strain tensor @;; for turbulent channel flows reads

®jj = —dypr(Cie + CiPoaj + ¢rprCle(aiar — 3A28;)

— @yorCa(Py — 3Piby) — ¢y prCa(Ej — 3Pidy)
— ¢y pp(0.033 + 0.6545 kS — C3 (I — 1850 + @) (5.1)

Here the near-wall correction tensor 4551’ is given by So et al. (1996)

q§l‘;} = fwdror[(Cre + CTPk)aij — C’le(aikakj — %Az@i;) + (X*(P,'j — %Pk&j) + 2)/*](51']'],
(5.2)

where f,, is a damping function given as f,, = exp[—(Re; /200)?], with Re; being the
turbulent Reynolds number Re; = k?/ev. Here o* and y* are constants in the SSG model,
which can provide a better prediction for a fully developed channel flow in the near-wall
region.

Comparison between the pressure—strain terms obtained directly from (2.9) and from the
model (5.1) using the DNS data for cases 1 and 3 is shown in figure 11. The model predicts
well the pressure—strain term for both shear-induced turbulence and particle-induced
turbulence in the bulk region, but the accuracy of the model is not good in the near-wall
region. It is a difficult issue how to accurately model the pressure—strain term in the
near-wall region for low-Re wall-bounded flows even for the single-phase case.

In the modelled ¢ equation, to account for near-wall effects, the term &, is modified as

€&
e = Ce2¢fpf_’ (5.3)
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Figure 11. Comparison between the pressure—strain terms @;; obtained directly from (2.9) and from the model
(5.1) using the DNS data for (a) (Re, =9.5) and (b) case 3 (Re, = 113.5). All values are normalized with

,ofuz/H.

where & = & — 2v(+/k/y)?, with y being the normal distance from the wall. The difference
between & and ¢ disappears at the edge of the viscous region for the low-Reynolds-number
case (e.g. Re; = 180) (Hanjali¢ & Launder 2011).

5.2. Comparison between RANS and DNS

We consider vertical channel flows, with gravity points downwards. We define
the streamwise, wall-normal and spanwise directions as x, y and z, respectively,
and also as x1, xp and x3, respectively. For fully developed channel flows, we
have (uju)) = (uju)) =0, (u2) = (u3) = 0 and d(u;)/dx; = d(u1)/dx3 = 0. Thus, the
dimensionless volume-averaged equations for the fluid phase can be simplified as

0 (ur) dp) 1 0 0{uy) (e () us))
= _— - — —F s
o Ui T Reom ¥ ony ax2
a¢f<u1u]> 0{uy)
— U — 2 (uuly) —— + D1y + @1y + e11 + I,
a7 /d’{(uluz) o2 1 1n+en+1In
0Qr (UsUty)
% = Dy + P + €20 + 12, (5.4)
Aoy (uyul) ’
—— = D33 + D33 + €33 + I33,
Ay (uyury) 0(uy)
% = —¢f<bt'2u/2)8—xz + D+ P12+ e12 + 112,
3¢f8
S =Pt Dot et

The pressure gradient is determined from the constraint that the flow rate (or bulk
velocity) is constant. The following boundary conditions are imposed on the wall: (u;) = 0,
(u;uj/.) =0ande = 2v(dﬂ/dx2)2. Equations (5.4) are solved with an in-house code using
a time advancement scheme to achieve the steady-state solution.

The full two-fluid RSM is complex (Fox 2014; Baker et al. 2020), involving some
sub-models for the solid phase, such as the lift model and the particle stress model. In
order to evaluate the performance of the interfacial term models proposed in the present
study, the uncertainty from other models should be minimized. Therefore, inspired by the
work of Ma et al. (2020b), we only solve the fluid-phase transport equations (5.4), and
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Cs ¢ () G G Cf Cy a* v* Cei  Ca o fs
0.22 1.7 1.05 0.4125 0.0 0.45 0.2125 —0.29 0.065 1.44 1.83 1.36 0.9

Table 6. Coefficients in the present RANS simulations of particle-laden channel flows.

directly use the DNS data for the solid-phase quantities involved in the fluid equations,
such as the particle volume fraction, the mean slip velocity and the mean drag. Table 6
presents the empirical coefficients in our RANS simulations.

The DNS mean drag is obtained from the fluid mean momentum equation in the
dimensional form at statistically steady state

d d dpe ’o
FD:@(-“ - >)+¢ ( § )+—(¢fpf( uyi43)).- (5.5)

From the domain-averaged momentum balance for the mixture, the mean pressure gradient
—dpe/dx =ty /H + ¢o(ps — /Of)g for upward flow, while —dp,/dx = 7,,/H — ¢o(ps —
pr)g for downward flow.

To evaluate the performance of the original near-wall RSM for the single-phase flow,
we compare the results on the single-phase turbulent channel flow from the model with
the DNS results of Hoyas & Jiménez (2008) in Appendix B. It is shown that the model
can predict well the turbulence statistics, which lays a foundation for the application of the
RSM to particle-laden flows.

The results on TKE from the RANS simulations of upward particle-laden turbulent
channel flows for cases 1-3 in table 5 are compared with the DNS results of Yu ef al. (2021)
in figure 12. Turbulence suppression across the entire channel is observed for cases 1
(Rep =9.5) and 2 (Re, = 49.6), whereas turbulence augmentation in the bulk region
takes place for case 3 (Re, = 113.5). The RANS results agree well with the DNS results,
as shown in figure 12. To the best of our knowledge, this is the first demonstration of
using RANS simulations to well predict particle-induced turbulence enhancement and
attenuation in vertical channel flows.

Figures 13 and 14 show that the root-mean-square (r.m.s.) values of the fluid velocity
fluctuations and some selected terms in the Reynolds stress equation of (i} u}) from RANS
agree well with the DNS results, which indicate that the particle-induced anisotropy is
correctly captured by our RSM. Nevertheless, significant over-estimation of v/, . and

W), near the wall by the RANS simulations can be observed in figure 13(a). We note
that such over-prediction occurs for the RANS simulation of the single-phase flow, as
shown in figure 22. In addition, figure 13 shows that the discrepancy of wall-normal and
spanwise r.m.s. velocities between Euler—Euler RANS and IR-DNS becomes relatively
larger in the centre region of the channel at a higher particle Reynolds number, which
is a surprising result, since one may assume that the flow tends to become a more
homogeneous sedimentation system as the particle Reynolds number increases. From our
previous study (Yu et al. 2021), the shear turbulence (characterized by the Reynolds shear
stress) in an upward channel flow can be attenuated by the settling particles, however,
the effect is non-monotonic with increasing particle Reynolds number. Above the critical
Rep, the Reynolds shear stress increases with increasing Re,, (see also figure 16), which
may indicate that the flow deviates more from the homogeneous sedimentation system at a
higher Re,,. Our models for the interfacial terms are based on the data of the homogeneous
sedimentation in a periodic domain, and we conjecture that this is the reason why the
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Figure 12. Profiles of TKE for (a) case 1 (upflow, Re, = 9.5), (b) case 2 (upflow, Re, = 49.6), (c) case 3
(upflow, Re, = 113.5).
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Figure 13. Root-mean-square velocity components from RANS and DNS for (a) case 2 (Re, = 49.6) and
(b) case 3 (Re, = 113.5).

discrepancy in the channel bulk region is higher for Re,=113.5 than that for Re,=49.6 in
figure 13. We note that there is also a more pronounced discrepancy in the Reynolds shear
stress for Re,=113.5 (figure 16). We leave the improvement of the model in the presence of
shear turbulence effects for the near-wall region and the r.m.s.velocities in the bulk region
to a future study.

From figures 15 and 16, the fluid mean velocities and Reynolds shear stress from the
RANS model also agree well with the DNS results. As Re), increases from 9.5 to 49.6,
the mean velocity profile becomes flatter and the Reynolds shear stress becomes much
smaller, which are well captured by the RANS simulations.
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Figure 14. Comparison between RANS and DNS on the selected terms in the equation of < u}u) > for
(a) Case 2 (Rep, = 49.6), and (b) Case 3 (Re, = 113.5). All terms are normalized with pfui /H.
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Figure 15. Mean fluid velocity profiles of turbulent channel flows for (a) case 1 (upflow, Re;, = 9.5),
(b) case 2 (upflow, Re, = 49.6) and (c) case 3 (upflow, Re, = 113.5).
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Figure 16. Reynolds shear stress —(u/v") profiles for (a) case 1 (upflow, Rep =9.5), (b) case 2 (upflow,
Re, = 49.6) and (c) case 3 (upflow, Re, = 113.5).

The profiles of TKE for cases 4-6 in table 5 are shown in figure 17. The results show
that the RSM also works well for the situations of higher flow Reynolds number, downward
flow and high density ratio. Particularly, figures 12(b) and 17(b) indicate that the effects of
flow direction on TKE can be captured by the RANS model: from figure 12(b), for upflow
at Re, = 49.6 (case 2), the turbulence is significantly suppressed by particles, whereas
from figure 17(b), for downflow at Re, = 38.3 (case 5), the turbulence is roughly not
attenuated by particles, particularly if compared with the DNS result of the single-phase
flow (Xia et al. 2021). Xia et al. (2021) attributed the reason for the difference to the
observations that, for case 2, the particles migrate towards the channel centre, and the
shear-induced large-scale vortices are significantly suppressed, whereas, for case 5, the
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Figure 17. Comparison of the TKE profile predicted by RANS and DNS for (a) case 4 (upflow, Re,=102.1,
Rep, = 12000), (b) case 5 (downflow, Re,=38.3) and (c) case 6 (downflow, p, = 100, Re,=90.6).
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Figure 18. (a) Mean fluid velocity, (b) Reynolds shear stress —(u'v’) and (¢) TKE profiles predicted for
case 9 (¢po = 14.2 %, Rep, = 22.9).

particles migrate towards the channel wall, and the flow is still dominated by large-scale
vortices. It is interesting that the RANS simulation does not resolve the vortices, but is
able to predict the turbulence attenuation or enhancement quantitatively.

The profiles of mean fluid velocity, Reynolds shear stress —(u'v’), as well as TKE for
case 9 (¢o = 14.2 %, Re,=22.9) in table 5 are shown in figure 18. The results show that
our model prediction is still good for larger volume fractions. We note that in the bulk
region the shear turbulence is almost completely suppressed by the particles in this case
(9o = 14.2 %, Re, = 22.9), evidenced by vanishingly small Reynolds shear stress.

In the following we attempt to identify the main reason why the appreciable turbulence
attenuation occurs in the RANS simulations for case 2, but not for case 5. Vreman (2015)
demonstrated that the non-uniform part of the drag force contributed significantly to the
particle-induced turbulence attenuation in point-particle simulations. The same analysis
method is used here, with a minor difference that the effect of the particle volume fraction
is taken into account, which was not considered by Vreman (2015) for point-particle
simulations.

The domain-averaged mean momentum equation of the fluid phase can be derived from
(5.5) as

—( dpe\ +—

where @ and Fp are the domain-averaged fluid volume fraction and drag force. If this
equation is subtracted from (5.5), we obtain

d a<u1>> o ( dpe)+w+’_w e (5.7)

&xwwx dx; & H P77
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Figure 19. Profiles of the terms defined in (5.8) for (a) case 2 (upflow, Re, = 49.6) and (b) case 5 (downflow,
Rep, = 38.3). All terms are normalized by ,ofu%.

where F}}' and (b}”‘ are the non-uniform part of the drag and the fluid volume fraction,
respectively, i.e. Fj)' = Fp — Fp and ¢f'~’” = ¢y — ¢. Integrating (5.7) from the wall
(x2 = 0) to xp = y, we obtain

;
nu

dx

y /Od’f 2

0 Rt iy

)7
(Brop(—uyup)) = Foolps —pp)g | &f" dx
— — 0

R

G
T
d{u Y
- <¢fﬂ ( 1>)+/ P dx, . (5.8)
dx 0
va F

Figure 19 shows the profiles of the terms in (5.8) for case 2 and case 5 obtained from the
DNS data of Yu ef al. (2021) and Xia et al. (2021). Since the particle volume fraction is
low, the term g is negligibly small. The drag force term tf is negative for both cases 2 and
5, which means that the drag tends to attenuate the fluid Reynolds stress, from (5.8). For
case 2, the interphase drag is larger in the channel centre region due to particle migration
towards the channel centre and thereby a higher particle volume fraction there (Yu et al.
2021), which leads to the integration term of the non-uniform force t7 having a larger
negative value in the bulk region, compared with case 5. Therefore, the non-uniform drag
distribution is responsible for the turbulence attenuation, as pointed out by Vreman (2015).
The turbulence enhancement at relatively high particle Reynolds numbers is clearly caused
by the TKE interfacial term I;. The accuracy of the interfacial term models is important to
the quantitative prediction of the turbulence attenuation and enhancement. One advantage
of the RSM over the eddy viscosity model for the multiphase flow is that the effect of
the interphase drag on the Reynolds shear stress can be adequately accounted for by the
former.

We also compare our RANS results on TKE and the r.m.s.velocities to the DNS results
of Uhlmann (2008) and Santarelli et al. (2016) in figure 20, and good agreement can be
seen.

952 A25-26


https://doi.org/10.1017/jfm.2022.927

https://doi.org/10.1017/jfm.2022.927 Published online by Cambridge University Press

(a) 0.05
0.04
0.03
0.02

0.01

(c) 04
0.3

0.2

r.m.s.

0.1

Figure 20. Comparison between the present RANS results, TKE and the r.m.s. velocities and the IR-DNS
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results of (a,c) Uhlmann (2008), case 7, and (b,d) Santarelli et al. (2016), case 8.

6. Conclusions

Based on data from the IR-DNS of particle sedimentation in a periodic domain at
a density ratio ranging from 0.01 to 1000, a particle concentration ranging from
2.3% to 30.2% and a particle Reynolds number below 250, the novel models for the
particle-induced interfacial terms for the second-moment closure have been developed.

The main conclusions are drawn as follows.

(1) The interfacial term in the dissipation rate equation is modelled as

I = Cg3

v
22
dp

Iy

6.1)

where I, is the interfacial term in the TKE equation modelled as Iy = Fpu,, and the
coefficient C.3 is modelled as

C d,F
= _ Re, rL . 6.2)
Ce2 (1 — @) o (k/ud )i
In (6.2) the relative TKE k/ uf is modelled as
k _
— =0.920; 0025078 1 0.05¢%3 (1 — ¢)(Re,/200) !
z
+25¢%(pr +0.5) 7051 — ¢p)P exp(—¢'2(1 — ) ?Re,'/?).  (6.3)
The correlation for the mean drag coefficient (Fp = Bu,) is as follows.
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For¢ < 0.1,
3 prdp(1—)u -
B = CH T (- ), (6.4)
4 d,
where Cy is the modified drag coefficient
Cy= L(l +0.15((1 — ¢)Re,) %87y (6.5)
(1 — @)Re, Y ’
and n is the exponent of the correction function in the form
n = (3.88 — 0.38In(Rep)) In(¢p ") p, 0038, (6.6)
For ¢ > 0.1,
2 1—¢)p(d1+15
8, — 180 2M¢ Lgtl=9) <¢>(2 + 1.5/9)
(1 — ¢) d?
18 (1 — ¢)Rey, [ 0.00456 (
+ 0.11¢(1 +¢) — ——— + [ 0.169(1 — ¢)
dy (1—¢)*
0.0644 B B >
——— ) Re 0343] +2.76Rey; p; 0 ————. 6.7
(1—¢)4> " w PR g 7

(ii) The particle-induced force term in the pressure—strain terms is modelled as @;; 3 =
—C3(I; — %Ikéij), with the coefficient C3 correlated with

C3(¢.Rep) = ¢ + aRe, + bRe, + c, (6.8)

inwhicha = 3.723 x 1074, b = 2.819 x 10~% and ¢ = —0.010. The interfacial term
in the Reynolds stress equation is modelled as I;j = Fp;u,j + Fpjuy;, and for the
sedimentation system, an only non-zero component is /1] = 2.

(iii) The anisotropic model for the dissipation rate tensor

()

J
k

2
&jj = ¢rpy (fs + §<1 _fv)sij> g, (6.9)

with f; = 0.9, is in excellent agreement with our DNS data on the particle-laden
turbulent channel flows.

(iv) The fluid-phase RSM for the particle-laden flows is established with the original
RSM for the single-phase flow and the above models for the particle-induced source
terms, and is found to be able to quantitatively predict the turbulence enhancement or
suppression by particles in vertical channel flows, through the comparison between
the RSM-based RANS simulations and the IR-DNS data in the literature. It is shown
that the non-uniform drag distribution is responsible for the turbulence attenuation
in the upward channel flow. We think that the capability to account for the effect of
the drag on the Reynolds stress is the advantage of the RSM over the eddy viscosity
model for the particle-laden flows.

Since our correlations are obtained from the sedimentation system, they are suited
to the case where the particle sedimentation effect (or the interphase mean slip
velocity) exist. For convenience, we assume that the coefficients in the single-phase
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dy/Ax =12.8 dy/Ax =25.6 dp/Ax =512

¢ =236% uy/ue 1.036 1.084 1.094
k/u? 0.059 0.062 0.063
¢=142% U /Ue 0.627 0.661 0.670
k/u? 0.225 0.219 0.223

Table 7. Effects of grid resolution on the mean relative velocity and TKE normalized with u. at Ga = 170,

L/d, =5and p, = 2. u. = \/4d,(p, — )g/3.

L/d,=5 L/d,=10 L/d, =20

Ga=173 uyfue 0.518 0.529 0.533
k/u? 0.215 0.231 0.233
Ga=170  uy/uc 0.661 0.697 0.714
k/u? 0.207 0.207 0.208

Table 8. Effects of the domain size on the mean relative velocity and TKE normalized with u, at ¢ = 14.2 %,
dy/Ax =25.6 and p, = 2.

RSM are not modified by the particles. Based on DNS data, extensive studies on
the full two-fluid RSM can be conducted in the future, such as the lift force model,
the solid-phase stress model and how the coefficients in the single-phase RSM are
modified by the particles.
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Appendix A. Box size and mesh resolution validations

The DF/FD solver has been extensively validated in our previous studies for the turbulent
channel flows (Wang et al. 2016; Yu et al. 2016, 2019, 2021) and particle sedimentation
(Yu & Shao 2007; Shao, Wu & Yu 2012; Xia et al. 2022b). It was shown that the
mesh resolution of d;,/ Ax = 12.8 is able to predict with reasonable accuracy the settling
velocity of a single particle at relatively low particle Reynolds numbers (e.g. Re, < 100)
(Shao et al. 2012; Xia et al. 2022b). Here we perform a mesh convergence test for the
sedimentation of many particles at relatively high particle Reynolds numbers. The time
step is determined from a fixed Courant—Friedrichs—Lewy (CFL) number CFL ~ 0.15,
where the CFLnumber is defined as CFL = u.At/Ax. The results of the mesh convergence

test on the mean relative slip u,/u. and the relative TKE k/ u% at Ga = 170 (Re,, ~ 210 for
¢ = 2.3 %) are presented in table 7, which shows that the relative differences in both
mean velocity and TKE obtained with d},/ Ax = 25.6 and 51.2 are below 1.5 %. Limited
by our computational resource, the mesh resolution of dj,/Ax = 25.6 is chosen for our
simulations.
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Figure 21. (a) Two-point correlation function of vertical fluid velocity as a function of vertical separation
distance for different domain sizes at ¢ = 14.2% and Ga = 73, (b) fluid velocity correlation functions for
Ga=73and 170 at ¢ = 14.2%, L/d, = 10 and p, = 2.
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Figure 22. Comparison between the near-wall RSM-based RANS simulations and the DNS results of Hoyas

& Jiménez (2008) for the particle-free turbulent channel flow at Re; = 180. (@) Mean velocity, (b) TKE,

(¢) Reynolds shear stress and (d) r.m.s. velocities. The lines and the symbols represent the DNS results of
Hoyas & Jiménez (2008) and the present RANS results, respectively.

In addition, we examine the dependence of the results on the domain size. The results
on the effects of the domain size on the mean relative velocity and TKE normalized with
ucat ¢ = 14.2 %, d,/ Ax = 25.6 and p, = 2 are presented in table 8, which indicates that
both mean relative velocity and TKE increase slightly with increasing domain size, and
they change little as the domain size is increased beyond L/d,, = 10.
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Figure 23. Comparison between the near-wall RSM-based RANS simulations and the DNS results of Hoyas
& Jiménez (2008) for the particle-free turbulent channel flow at Re; = 395. (a) Mean velocity, (b) TKE,
(c) Reynolds shear stress and (d) r.m.s. velocities. The lines and the symbols represent the DNS results of
Hoyas & Jiménez (2008) and the present RANS results, respectively.

The two-point autocorrelation function for the vertical component of the fluid
fluctuating velocity as a function of vertical separation distance can be computed from

Ruu(rx) — (M/(xvyv Z7 t)u/(x+rx’y’ Z’ t)) (A])

(W (x,y,2,1)%)

Figure 21(a) shows the velocity correlation functions for different domain sizes at Ga =
73 (Rep ~ 44) and ¢ = 14.2 %. The results show that the two-point correlations vanish
at the separation of half-domain height for L/d,, = 10 and 20. The velocity correlation
function for Ga = 170 (Re, ~ 137) and ¢ = 14.2 % at L/d,, = 10 is shown in figure 21(b),
and no correlation is observed at the separation of half-domain height, unlike the dilute
case where strong clustering of particles in the form of vertically elongated columnar
regions were observed by Uhlmann & Doychev (2014) at Ga = 178. The results on the
two-point autocorrelation function and the settling velocity above indicate that a domain
size of L/d, = 10 is sufficiently large to ensure a rough independence of the results on
the domain size. Nevertheless, some results for L/d,, = 5 (denoted as the ‘small’ case) are
also reported, as compared with those for L/d,, = 10 (denoted as the ‘large’ case), since
the small size of the computational cell may also be used for two-fluid RANS simulations.
Unless otherwise specified, the results reported are obtained with L/d), = 10.
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Appendix B. RANS simulations of fully developed particle-free turbulent channel
flows

The test cases are the particle-free turbulent channel flows at Re; = 180 and 395. Here
Re: is the friction Reynolds number defined by Re; = uH /v (u; = 1,/ oy being the wall
friction velocity and t,, being the mean shear stress on the walls).

Profiles of the fluid mean velocity, TKE, Reynolds shear stress and r.m.s. velocities are
compared with the benchmark DNS data of Hoyas & Jiménez (2008) in figures 22 and 23.
From these figures, our results obtained from the near-wall RSM are in good agreement
with those of Hoyas & Jiménez (2008).
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